Author’s response to reviews

Title: Improving Nutrition and Physical Activity Environments of Family Child Care Homes: The rationale, design and study protocol of the ‘Healthy Start/Comienzos Sanos’ cluster randomized trial

Authors:
Patricia M. Risica (Patricia_Risica@brown.edu)
Alison Tovar (Alison_Tovar@uri.edu)
Vanessa Palomo (Vanessa_Palomo@brown.edu)
Laura Dionne (laura_dionne@brown.edu)
Noereem Mena (mena23@my.uri.edu)
Kate Magid (kate_magid@alumni.brown.edu)
Diane Stanton Ward (dsward@email.unc.edu)
Kim Gans (Kim.Gans@uconn.edu)

Version: 2 Date: 21 Feb 2019

Author’s response to reviews:

The revisions requested in the most recent email were addressed as follows:
1 - trial site date was updated per request
2 - cover page was omitted
3 - final versions of all documents, tables and figures were submitted.